
○ Directorate wants to gather interest on using the Tevatron for accelerator physics experiments.

○ January 13-14, 2010 will have a workshop at Fermilab

� ronmoore@fnal.gov

� X6466

○ Let Ron know if you want to attend.

• Ron Moore:  Guest speaker

○

30

0

0 10

A:STCKRT

M:TOR109

10 Weeks

○ Rolls off above 8e12

• Regular Meeting

2009-12-17 Pbar Department Meeting
Thursday, December 17, 2009

10:01 AM

   Meetings Page 1    



○

30

0

0 80

A:STCKRT

Sep 22-Nov30

Dec 1- Dec 17

D:TGTPWR

○ Try target power.  Things are better.
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○

A:STCKRT/D:TGTPWR

Sep 22 - Dec 17

D:TGTPWR

A:STCKRT

○

○ Ready for study

○ Have to set the gains locally

○ Have to restart SA to restart gain.

• SEM status

○ Want consistent beam and consistent TLG

○ Start looking at coming in during the evening

• Proton Spot Size

○ Jim Budlong has Java code ready to test new hardware
• SEM806 (new vs old hardware; timer)

○ General tuning.

○ Have a new aggregate "tight Debuncher Apertures"

• Debuncher Apertures

○ A few systems have had trips
• DRF1

○ Still oscillating
• Pbar Frig

○ Broken.  Very little power.  Downstairs problem.
• Deb HB1

○ Original problem.  Module #2 not firing, bad trigger module

○ Turned PS off, then couldn't turn it back

○ Output of replacement low

○ Trigger problem fixed

○ Put original Spellman

○ Took front cover, wire next to circuit board shorting board when button pushed.

○ Original spellman back in.

� Max V without load
� Max I shorted

○ Test without a load

○ May be good, but my have had the current pot set too low.

• Deb EKIK

� #3 goes on and off.  
� #30 off all the time
� #24 almost dead.  No pad, only puts out 26-28 watts.  
� #19 pulled enough padding out to get normal power.

○ TWT03, TWT30, TWT19, TWT24
• Stacktail

Stacking

• Topics
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� #19 pulled enough padding out to get normal power.
� #19 and #24 was old delta kicker.  No used, but were on, so they aged.
� We have 6 good spares….90 day turn around…kluge since they don't make that 

tube anymore.

� Three settings used to work.
� Those settings don't work anymore.

○ Filters

� #19 was a wavelength off

○ Fanbacks

○ Monitor

� Was actually the readback.
� Wes swapped out with a spare.
� Still using old style temp controller as an analog readback.

○ Oven Temp

� Getting better.
� No more long term excursions
� Short excursions of small amplitude only

○ CHAOS

� Different electronics for BPM
� Want an alternative standalone BPM
� Attached to BPM514

○ Radial Feedback?

• A3B7

○

○ DS end not pointing toward C-magnet 

○ 4cm move into aisle would line it up

○ V901 was a tight aperture

○ The desired position changed a couple mm.

• ELAM Position

Transfers

○ Down to two vehicals

○
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